Author's response to reviews

Title: Status of HBV infection and vaccination among health care workers in a Public General Hospital: a retrospective cohort study

Authors:

Yu-bao Zheng (guangzhouzyb@126.com)
Yu-jie Su (suyujie1987@hotmail.com)
Ke Wang (wangke19821@126.com)
Yu-rong Gu (865576465@qq.com)
Zhan-lian Huang (zhanlianh@21cn.com)
Chao-shaung Lin (shuangss@21cn.com)
Zhi-liang Gao (zhilianggao@21cn.com)

Version: 2
Date: 3 November 2014

Author's response to reviews:

Dear Editor,

We are very pleased to learn from your letter about revision for my manuscript entitled “Status of HBV infection and vaccination among health care workers in a Public General Hospital: a retrospective cohort study” (Ms. No. Manuscript ID MS: 2011913633145142).

We wish to thank you for your attention and the editor for the constructive comments and advice. We have revised the manuscript according to the comments from you.

We have modified Title Page: Title Page including all the co-authors' email addresses in the Title page and Ethics’ statement. And the Line Numbering and revise our manuscript to include line and page numbers(as our revised version of the manuscript).

A list of abbreviations used in the manuscript and their meanings after the Conclusions section.

With many thanks for your cordially help and patience, we are looking forward to your positive response as soon as.

With kind regards,

Yu-bao Zheng, Associate Professor. MD,
Department of Infectious Diseases, The Third Affiliated Hospital of Sun Yat-Sen University, No. 600 Tianhe Road, Guangzhou 510630, Guangdong Province, China.
Telephone: +86-20-85252250 Fax: +86-20-85252250
E-mail: guangzhouzyb@126.com.